Author's response to reviews

Title: High frequency of Machado-Joseph disease identified in Southeastern Chinese kindreds with spinocerebellar ataxia

Authors:

Shi-Rui Gan (ganshirui1981@163.com)
Sheng-Sheng Shi (shishengsheng@gmail.com)
Jian-Jun Wu (jungliw@gmail.com)
Ning Wang (nwang63@yahoo.com.cn)
Gui-Xian Zhao (zhaogx5639@yahoo.com.cn)
Sheng-Tong Wen (wst3344@126.com)
Shen-Xing Murong (zhiyingwucn@yahoo.com.cn)
Chuan-Zhen Lu (luchuanzhen@yahoo.com.cn)
Zhi-Ying Wu (zhiyingwu67@yahoo.com)

Version: 5 Date: 16 March 2010

Author's response to reviews: see over
Cover Letter

To the editor:

We have made all formatting changes to the format of our manuscript requested by editorial production team.

**Major Revisions:**

1. Tables: We have included the tables in the manuscript in a tables section following the references.
2. References: Reference 60 has been cited in the Discussion section.

**Minor revisions**

All changes have been made according to your suggestions.

Sincerely yours,

Zhi-Ying Wu, M.D.&Ph.D.